**Author's response to reviews**

**Title:** Clinical features and gene mutational spectrum of CDKL5-related diseases in a cohort of Chinese patients

**Authors:**

Zhao Ying (zhaoying2008renji@163.com)
Zhang Xiaoying (zxy269163@163.com)
Bao Xinhua (zwhang@pku.edu.cn)
Zhang Qingping (zhangqqp1988@126.com)
Zhang Jingjing (zjj25952@163.com)
Cao Guangna (dudu.laile@163.com)
Zhang Jie (bingling8766@126.com)
Li Jiarui (lijr@mail.cbi.pku.edu.cn)
Wei Liping (weilp@mail.cbi.pku.edu.cn)
Pan Hong (Panmuren@263.com)
Wu Xiru (wwning@bbn.cn)

**Version:** 4  
**Date:** 10 February 2014

**Author's response to reviews:**

1. We have added emails of all authors in the title page

2. We have changed the section title “Conflict of Interest” to “Competing Interests”

3. We have provided the full name of the hospital research ethics board in the “Methods” section/ “Patients” sub-section